#### DOCUMENT RESUME

ED 267 276

CE 043 949

TITLE

Literature Review on Improving Secondary Vocational

Education Effectiveness.

INSTITUTION

Northwest Regional Educational Lab., Portland, OR.

Education and Work Program.

SPONS AGENCY

Office of Educational Research and Improvement (ED),

Washington, DC.

FUB DATE

Mar 86

CONTRACT

400-86-0006

NOIE

48p.

The second secon

PUB TYPE

Information Analyses (070)

EDRS PRICE

MF01/PC02 Plus Postage.

**DESCRIPTORS** 

Academic Education; \*Administrative Principles; Business Administration; \*Educational Improvement; \*Educational Policy; \*Educational Research; School

Districts; \*School Effectiveness; Secondary

Education; \*Vocational Education

#### **ABSTRACT**

The current demand for educational excellence is an outgrowth of recent national reports criticizing secondary education. This push for academic excellence also affects vocational education outcomes. A growing consensus seems to be emerging that at the secondary level (1) vocational education should be integrated with academic education, (2) technical skills should focus on transferability, (3) employability skills are important, and (4) basic communication, computation, and problem-solving skills need reinforcement in occupational preparation. Principles derived from effective schooling research, successful business practices, and school improvement policy studies can help improve vocational education. Effective schooling studies have identified six critical factors with the greatest potential for application to vocational education settings: time on task, expectations of performance, student motivation, cooperative learning, effective learning climate, and the use of evaluation. The eight successful business practices described by Peters and Waterman in "In Search of Excellence" are also relevant to vocational education: a bias for action, close to the customer, autonomy and entrepreneurship, productivity through people, hands on -- value driven, stick to the knitting, simple form--lean staff, and simultaneous loose-tight properties. Finally, school improvement policy studies focusing on the co-management of the school by the school and the district show that district support can be a key factor in improving vocational education. A seven-page bibliography is included. (SK)

Michigan Committee a Basel Season Season See at



#### LITERATURE REVIEW ON

#### IMPROVING SECONDARY VOCATIONAL EDUCATION EFFECTIVENESS

#### Proposed by

The Education and Work Program
Northwest Regional Educational Laboratory
300 SW Sixth Avenue
Portland, OR 97204
(503) 248-6800

#### U.S. DEPARTMENT OF EDUCATION

NATIONAL INSTITUTE OF EDUCATION EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC)

- hs document has been reproduced as received from the person or organization originating it.
- [] Minor changes have been made to improve reproduction quality
- Points of view or opinions stated in this document do not necessarily represent official NIE position or policy.

#### March, 1986

Sponsored by Office of Educational Research and Improvement

This publication is based on work sponsored wholly, or in part, by the Office of Educational Research and Improvement (OERI), Department of Education, under Contract Number 400-86-0006. The content of this publication does not necessarily reflect the views of OERI, the Department, or any other agency of the U.S. Government.



#### I. INTRODUCTION

#### Purpose

This literature review is being conducted as background for a NWREL study on improving secondary vocational education effectiveness. The literature review draws heavily on national studies and reports on effective schooling, principles of successful business operations, and policy studies related to school improvement. This preliminary literature review will be examined by several researchers and policy makers and expanded based on their input.

#### Audiences

As a result of recent national reports critical of secondary education in the United States, increased attention is being given at the local, state, and national level for the need to develop academic excellence in education. In the haste to define academic excellence, many policy makers and educators have ignored the question of what vocational excellence means and how to achieve it. This paper takes an exploratory look at vocational excellence and suggests how principles derived from: 1) effective schooling research, 2) successful business practices, and 3) school improvement policy studies could be used to enhance our understanding and improvement of vocational education. This literature review is intended for interested vocational education policy makers, vocational directors, legislators, advisory committees, vocational instructors, employers, and those involved in establishing and carrying out an agenda for the future research in vocational education.

#### Description of Contents

Section II of this paper discusses new demands for educational excellence in light of recent national reports criticizing secondary education. Implications of the push for academic excellence for vocational education are described together with efforts to identify expectations for vocational education.

The main focus of this paper is illustrated in Chart I which displays selected prinicples from effective schooling research, successful business practices, and school improvement policies. Section III of this paper discusses the origins of effective schooling research and reviews selected findings from a macro and micro perspective. Limitations of effective schooling research are listed. Ways in which effective schooling research has been linked to school improvement are described. Also described in Section III are ways in which findings from effective schooling research can be applied to secondary vocational education. Section IV focuses on the application of findings from studies of successful businesses. While Section V describes the literature and applications from school improvement policy studies research. Section VI contains the bibliography.



-1- 3

#### TABLE 1

# APPLICATIONS OF SELECTED PRINCIPLES FROM EFFECTIVE SCHOOLING RESEARCH, SUCCESSFUL BUINESS PRACTICES, AND SCHOOL IMPROVEMENT POLICIES TO SECONDARY VOCATIONAL EDUCATION

Effective	Schooling
Practices	

- O Time on Task
- o Expectations of Performance
- o Student Motivation
- o Cooperative Learning
- O Effective Learning Climate
- o ·Use of Evaluation

#### Successful Business Practices

- O A Bias for Action
- o Close to the Customer
- O Autonomy and Entrepreneurship
- o Productivity Through People
- o Hands On, Value Driven
- o Stick to the Knitting
- O Simple Form, Lean Staff
- O Simultaneous Loose-Tight Properties

## School Improvement Policies

- o District Policies
- O School Plans or Policies
- o District Support

### Applications to Secondary Vocational Education

- o Policy
- o Practice
- o Research



#### II. NEW DEMANDS FOR EDUCATIONAL EXCELLENCE

### Recent National Reports Criticize Secondary Education

When I say excellence, I am not referring to the top academic performers in our schools. By excellence I mean that every youngster should have the opportunity to succeed to her or his highest potential. That means excellence in our programs for the handicapped as well as the talented and gifted, in vocational education and the arts as well as in math and science.

The preceding quote by Oregon State Superintendent, Verne A. Duncan, in his annual; State of the Schools address (1983), makes a specific and deliberate plea for enhancing school improvement efforts so all students might achieve excellence. In a speech delivered to the Vocational Industrial Clubs of America (VICA) National Leadership Conference (1983), President Reagan emphasizes:

And each generation must realize that to achieve America's potential we need all our people with all their talents working together. That's why our drive for excellence in education must reach every student in every school in every subject. We should see that all our young people get a good grounding in English and literature, history, math, science and the other basics. But we must also recognize that our vocational classrooms are just as important as any other. And we should insist that the vocational courses we teach prepare this generation with the skills they need for real jobs.

However, in reviewing some recent national reports on the state of education today, we do not find that same commitment and appeal for equality for those students outside the college-bound stream.

Several recent reports criticizing American public secondary education have received national attention. The National Commission on Excellence in Education was charged with examining the quality of education in the United States and subsequently, to make a report to the Nation and to the Secretary of Education. The ensuing report, A Nation at Risk: The Imperative for Educational Reform (Commission Report 1983), addressed the following:



-3-

- O Assessing the quality of teaching and learning in our nation's public and private schools, colleges and universities
- Comparing American schools and colleges with those of other advanced nations
- Studying the relationship between college admissions requirements and student achievement in high school
- O Identifying educational programs which result in notable student success in college
- Assessing the degree to which major social and educational changes in the last quarter century have affected student achievement
- O Defining problems which must be faced and overcome if we are to successfully pursue the course of excellence in education

Notably absent in this list is equal attention given to the needs of noncollege bound students and students choosing vocational education. Though the report makes mention of those students not typically involved in college preparatory programs, minimal consideration is given to vocational education as a viable option for these students as evidenced in the following paragraph (Commission Report 1983):

We must emphasize that the variety of student aspirations, abilities, and preparation requires that appropriate content be available to satisfy diverse needs. Attention must be directed both to the nature of the content available and to the needs of particular learners. The most gifted students, for example, may need a curriculum enriched and accelerated bayond even the needs of other students of high ability. Similarly, educationally disadvantaged students may require special curriculum materials, smaller classes, or individual tutoring to help them master the material presented. Nevertheless, there remains a common expectation: we must demand the best effort and performance from all students, whether they are gifted or less able, affluent or disadvantaged, whether destined for college, the farm, or industry.

A second report to achieve widespread press was compiled by the Task Force on Education for Economic Growth: <u>Action for Excellence</u>. A Comprehensive Plan to Improve Our Nation's Schools. Eight Action Recommendations are proposed by the committee (Task Force Report 1983):

 Develop--and put into effect as promptly as possible--state plans for improving education in the public schools from kindergarten through grade 12.



- 2. Create broader and more effective partnerships for improving education in the states and communities of the nation.
- Marshal the resources which are essential for improving the public schools.
- 4. Express a new and higher regard for teachers.
- 5. Make the academic experience more intense and more productive.
- 6. Provide quality assurance in education.
- 7. Improve leadership and management in the schools.
- 8. Serve better those students who are now unrerved or underserved.

On the surface, Action Recomme: lation 8 would seem to address the needs of students not pursuing college preparatory studies. On closer inspection, however, the recommendation more directly speaks to the needs of equality in education for young women, minorities, and the handicapped.

Yet a third report to gain extensive public attention is the Carnegie report offering a plan for high school reform. High School: A Report on Secondary Education in America, a study by the Carnegie Foundation for the Advancement of Teaching, presents an agenda for action to improve our nation's secondary schools. The report identifies twelve themes that provide a framework for reform (Boyer 1983):

- 1. Clarifying Goals
- 2. The Centrality of Language
- 3. The Curriculum as a Core
- 4. Transition to Work and Learning
- 5. Service: The New Carnegie Unit
- 6. Teachers: Renewing the Profession
- 7. Instruction: A Time for Learning
- 8. Technology: Extending the Teacher's Reach
- 9. Plexibility: Patterns to Fit Purpose
- 10. The Principal as Leader
- 11. Connections
- 12. Excellence: The Public Commitment

The fourth theme, <u>Transition to Work and Learning</u>, gives credence to the importance of preparing students for the changing nature of the world of work (Boyer 1983):

The high school should help all students move with confidence from school to work and further education. Today, we track students into programs for those who "think" and those who "work," when in fact, life for all of us is a blend of both. Looking to the year 2000 we conclude that, for most students, 12 years of



schooling will be insufficient. Today's graduates will change jobs several times. New skills will be required, new citizenship obligations will be confronted. Of necessity, education will be lifelong.

As with the two preceding reports, this one gives no specific and direct solutions for preparing noncollege bound students for future work.

In reviewing national reports on the state of education today, one can easily discern the lack of equitable consideration for the needs of students wishing to pursue vocational programs. The push for academic excellence appears to leave out vocational education. Gordon Swanson (1983) in commenting on these national reports expressed concern that they confused means (e.g., more course requirements) with ends (multidifferentiated outcomes). He felt three essential questions need to be addressed by policy makers in discussing excellence in education—what should be taught, to whom, and for what purpose? The national reports on excellence focused on the first question but ignored the most important second and third questions.

In contrast to these reports on secondary education, the National Research Council Committee on Vocational Education and Economic Development in Depressed Areas has prepared a comprehensive report on the role vocational education should play in secondary education. In their book Education for Tomorrow's Jobs (Sherman 1983) the committee states:

Vocational education is a vital part of the public education system in this country, one that has long been slighted in favor of academic education. Basic academic as well as occupational skills are of fundamental importance in preparing young people for productive lives in our society. The public elementary and secondary schools in this country should offer students who will not go to college a thorough grounding not only in language skills, reasoning, and mathematics, but also in the mechanical and technical skills and work habits that will prepare them for working life (pg. 1).

The Committee for Economic Development in 1985 issued a report

Investing in Our Children: Business and the Public Schools. Part of the warning signs they site in education reflect employer dissatisfaction. They say that:

Employers in both large and small businesses decry the lack of preparation for work among the nation's high school graduates. Too many students lack reading, writing, and mathematics skills, positive attitudes toward work, and appropriate behavior on the job. Nor have they learned how to learn, how to solve problems,



-6-

make decisions, or set priorities. Many high school graduates are virtually unemployable, even at today's minimum wage (pg. 2).

A Nation at Risk and many of the other studies of the 1970s critical of public education focused on academic success and basically ignored the noncollege bound student. Investing in Our Children specifically focuses on the needs of all students including those not going on to college. This committee recommends that the term "vocational education" be restricted to specific preparation of students to enter a field upon graduation. They then advocate that students be required to demonstrate achievement of an adequate level of academic competence before completing occupationally specific training. They also recommend close business linkages with education.

#### Academic Excellence and Vocational Education

Some argue that the push to achieve academic excellence by increasing high school graduation requirements will benefit all students, that is, students planning to enter the work force immediately after graduation and those planning to enter and complete college. Since many high school students, including those who enter jobs after graduation, may eventually enter college, a broad background that would prepare them for that possibility is essential.

Others would argue that by raising the number of academic courses needed for high school graduation, students will not have time to gain the concrete training they need for job entry upon completing high school. This group feels that simply mandating more units of science and math for all students does not ensure that these courses will be of excellent quality. Those supporting the role of vocational education in the comprehensive high school share the belief that science, math, and communication skills are important. They argue, however, that for many students who are more action-oriented in their approach to learning, integrating math, science, and communication skills into applied vocational programs is a more appropriate way to help students master these skills.

Gene Bottoms, former executive director of the American Vocational Association, has observed:

The agenda for excellence that is rolling across America is far too narrow. It ignores the fact that individuals learn differently, come from different family backgrounds, and have different aspirations. It assumes that there is a single best way. It seems to be saying that doing more of what we are already doing badly will produce excellence (1983, pg. 9).



Regardless of the position one takes on the number of units of academic or vocational classes that students should be required to take, there is general agreement that the quality of all these courses and of education in general needs improvement. Thus, the focus of this paper is to address how factors that have been found successful in improving academic education and business can be applied to improving secondary vocational education. Since one important outcome of vocational education is to prepare students for job entry, we cannot understand what excellence in vocational education means without considering first what employers expect of newly employed high school graduates.

#### Employer Expectations

The Center for public Resources surveyed educators and employers in .1982 for their views on basic skills deficiencies among high school graduates entering the work force. The results revealed that "there is a considerable gap between the perceptions of school officials and the perceptions of employers when it comes to the adequacy of education for employment" (Task Force Report 1983). Employers were more critical of basic skills deficiencies.

Future workers will face complex problems. To perform successfully in the marketplace, workers will need to be resilient, versatile, independent, and able to interact cooperatively and ethically with others. A recent study conducted by the Northwest Regional Educational Laboratory (NWREL) found that employers in three major Northwest industries-electronics, wood products, and the health field-mentioned the following skills as important: communication skills, interpersonal skills, and the foundation of a solid but broad fundamental education (Crohn 1983). Adaptability is becoming increasingly important as a trait employers desire. "Things have changed so much and so rapidly," says one personnel manager, "that one thing we have to be sensitive about is the willingness of a person to accept change in the environment" (Crohn 1983). Similarly, a person who comes to new employment with the ability to perform more than one task is considered to be more valuable than the worker with a limite's skill range. Being "flexible" was a term heard in the NWPEL interviews with employers as a desirable attribute because of the potentially changeable nature of the work environment. This generally meant the ability to handle more than one task at a time.

Another study (Richards 1981) synthesized 11 earlier studies of employer expectations for young employees. Expectations centered on: basic academic skills, communication skills, knowledge of the world of work, interaction with fellow workers and superiors, positive attitudes toward work, dependability, craftsmanship and productivity.

In looking at employer attitudes toward and perceptions of the deficciencies in the job performance of young people, one study of over 800 employers uncovered three basic problem areas: 1) poor performance in basic skills, particularly oral and written communications; 2) poor work attitudes; and 3) a general lack of understanding about the world of business (Owens 1983).



-8-

Psychological and social maturity was found to be one of five factors that make youth more employable in a study done at the University of Minnesota (Hadin 1983). Key dimensions of maturity include a sense of social and personal responsibility, self-control, self-direction, and a sense of obligation to complete tasks. These findings are supported by a study at Par West Laboratory: "What employers want from the schools is motivated people with the understanding and maturity to fit into an adult organization, and the basic and fuctional skills that enable them to grow. Without these qualities, technical job skills are of limited value to the young worker, the employer, and to the nation's long-range needs for a more productive work force" (Peterson 1982).

## Identifying Vocational Excellence

Identifying excellence in vocational education demands an understanding of what vocational education is intended to accomplish.

Defining the purposes and expected outcomes of vocational education has been one of the biggest obstacles in planning and evaluating secondary vocational education. Although numerous studies of secondary vocational education have occurred, no consensus exists as to the key aspects that should be measured (Darcy 1980). Various approaches have been used to determine effectiveness of vocational education programs.

Interviews with over 80 vocational educators, advisory council members, and school administrators in the metropolitan Portland area revealed five expectations for secondary vocational education: specific job skills preparation, broader occupational skills, personal/social development, improvement of basic skills, and exployability skills development (McClure and Owens, 1986).

More than 100 vocational education outcomes have been identified in a Thesaurus of Outcome Questions (Farley 1979). These have been organized in various ways. For example, Harry Silberman has distinguished between intrinsic and extrinsic purposes (1983): "The intrinsic purposes are the immediate consumption values, regarded as ends in themselves: e.g., stimulation, recreation, understanding. The extrinsic purposes are the utilitarian values of the program, regarded as means to attain future benefits: e.g., employment, income, labor supply, economic productivity" (pg. 13). Silberman goes on to argue that vocational education has more direct influence over the intrinsic than the extrinsic outcomes. He identifies three intrinsic outcomes of concern to teachers of secondary vocational education: technical skills and knowledge, communication skills and literacy, and personal skills and attitudes.

In designing a statewide study of secondary vocational education effectiveness for Oregon, NWREL adapted a framework used at the National Center for Research in Vocational Education (NCRVE). Outcomes are organized under four headings—occupational, educational, ancillary, and societal (Owens 1982).



-9-

Examples under these categories are shown in Table 2. After reviewing much of the prior research on vocational education effectiveness, the NWREL researchers extracted a list of outcomes and of quality program indicators. A state project steering committee composed of vocational educators, administrators, and employers reviewed the lists of 20 outcomes and 20 indicators and provided a rating of importance for each outcome and quality program indicator. A similar process was used to obtain ratings from the vocational education research directors in Arizona, Florida, Illinois, Ohio, Oregon, and South Carolina. The rank orderings of these two groups on the outcomes were used as the basis for developing student, parent, employer, and vocational educator evaluation instruments.

The outcomes ranked most important were possession of occupational skills needed for job entry level, ability to apply basic literacy skills to work tasks, ability to meet the local labor market demand, interpersonal skills, pride in producing quality work, possession of good work habits, and development of transferable occupational skills. Quality program indicators receiving top priority were availability of instructional staff with adequate training and experience in their vocational area; improved linkage with business, industry, and labor; placement of completers in employment related to their vocational training; availability of quality cooperative work-experience opportunities; and realistic instructional experiences.

TABLE 2

## EXAMPLES OF SECONDARY VOCATIONAL EDUCATION OUTCOMES

Category	Illustration
Occupational	o Possession of occupational skills needed for job entry level
	o Possession of transferable occupational skills
Educational	O Adequate understanding of fundamental principles as a basis for advanced training
	o Articulation of high school and postsecondary training programs
Ancillary	o Possession of leadership skills
	o Ability to cope with changing technology
Societal .	o Providing a supply of people trained in emerging occupational areas
	o Effective collaboration between the business community and the schools



The Division of Occupational Education Programs in the New York State Education Department has been involved in a massive Occupational and Practical Arts Futuring Project to examine the future direction of vocational education in their state. Part of the process involved the use of a delphi technique to obtain consensus from many education committees and business representatives within New York regarding skills and knowledge to be developed through occupational and practical arts education. Their priorities, in rank order, are shown in Table 3. They also rated services of Occupational and Practical Arts Education to business and society. These priorities, in rank order, are shown in Table 4.

#### TABLE 3

#### RANK ORDER OF PROPOSED VOCATIONAL EDUCATION SKILLS AND KNOWLEDGE OUTCOMES RATED BY EDUCATORS AND BUSINESS PROPLE IN NEW YORK STATE

- 1. Basic technical skills which are common to a cluster of jobs
- Employability skills, such as promptness, dependability, and selfreliance
- Abilities in problem solving, communications, decision making, interpersonal relationships, and resource management
- 4. Basic skills in use of common hand tools and machines as well as understanding of various materials and work processes
- 5. Technological literacy (i.e., conceptual understanding of computers, communication systems, etc.)
- 6. Ability to cope with life situations and to realize self-awareness and self-worth
- 7. Technical reading, writing, and mathematics skills which relate to a specific occupational or practical arts instructional program area
- 8. Career awareness and the ability to make career and educational choices
- 9. Basic reading, writing, and mathematics skills
- 10. An awareness of, and positive attitude toward safety procedures
- 11. Knowledge of basic economic concepts
- 12. An ability to take advantage of inevitable change
- 13. Technical skills which are specific to one job
- 14. Awareness of the role and responsibility of individuals working alone and in groups
- 15. Ability to manage a home and personal business affairs
- 16. The capacity for self-fulfilling use of leisure time



-13-

#### TABLE 4

## RANK ORDER OF PROPOSED VOCATIONAL EDUCATION SERVICES RATED BY EDUCATORS AND BUSINESS PROPLE IN NEW YORK STATE

- 1. Be responsive to the needs of emerging occupations
- 2. Produce a supply of qualified graduates adequate to meet the employment needs of business and industry in the State
- 3. Provide program offerings which are based upon employment opportunities in the region or State, and not simply on student interest
- 4. Provide services which assist the student in making the transition from school to employment and/or to more advanced education
- 5. Provide accessibility to programs for all interested individuals
- 6. Provide training or retraining for unemployed/underemployed
- 7. Promote the right of every individual to a chosen career without regard to traditional barriers
- 8. Coordinate with general and postsecondary education and with military and industrial training programs
- 9. Meet short-term training needs of business and industry
- Be responsive to changing management structure (i.e., shared decision making and quality circles)
- 11. Be designed to improve the ability of individuals to be effective entrepreneurs
- 12. Develop an understanding of an individual's responsibility in protecting the environment for social and personal benefit
- 13. Promote and support the values of the American economic system
- 14. Contribute to the maintenance of a trained labor force for national defense needs
- 15. Develop an understanding of all economic systems



16

Recently, the federal Education Department established the Secretary's Award for Outstanding Vocational Education Programs. Criteria used in selecting these exemplary programs consisted of the following: 1) the presence of explicitly stated objectives consistent with the program purposes, 2) evidence of progress toward achieving those objectives, 3) the quality of instructional materials used by instructors and students, 4) appropriateness of equipment and facilities, 5) hands-on experiences for students, 6) planning and cooperation with local businesses, labor leaders, and other organizations, 7) use of evaluation and feedback, 8) high placement rate and employer satisfaction with the graduates, 9) cost-effectiveness, and 10) program recognition and/or publicity.

In summary, there is no national agreement regarding key outcomes of secondary vocational education. However, in our opinion, a growing consensus seems to be emerging that at the secondary level:

1) vocational education should be integrated with academic education,

2) technical skills should focus more on those common to a cluster of jobs than on a single occupation;

3) employability skills such as dependability, the ability to get along with others, and an understanding of the world of work are important; and 4) basic communication, computation, and problem solving skills should be reinforced as they are applied to occupational preparation. In the next section of this paper, we turn to effective schooling research as one base for improving vocational education.



## III. EFFECTIVE SCHOOLING RESEARCH OFFERS DIRECTION

#### FOR VOCATIONAL EDUCATION

#### Origins of Effective Schooling Research

An early influential school effectiveness study sponsored by the United States government was reported by Coleman and his colleagues in 1966. Coleman's nationwide survey of schools resulted in Equality of Educational Opportunity. A major conclusion of this report stated, "school brings little to bear in a child's achievement that is independent of his background and general social context" (Coleman 1966).

Coleman's conclusion came as a surprise; up until that point, American society held the common belief that schools are fundamental in reducing inequality. Another Coleman finding startled educators—school characteristics and resources seemed to make little difference in student measured levels of achievement. According to Coleman, home background prior to entering school was the most critical factor influencing student achievement.

The Coleman Report had major impacts on later school effectiveness studies. It not only accelerated the growing debate on the effectiveness of schools, it also altered perceptions as to what constitutes equal opportunity for students. Prior to the report, equal opportunity generally meant equal access to such resources as libraries, books, lab facilities, and so on. Coleman showed these to have relatively little effect on student achievement; the result was a dramatic shift in beliefs regarding equal education opportunity. Educators began to put the emphasis on tested achievement as a means of determining equal opportunity for students.

Reactions to the Coleman Report included some well-grounded criticisms ranging from procedural and mechanical errors in the ways the data were handled to defects in the analysis. However, it was the Coleman Report that began the serious inquiry into what constitutes effective schools.

Review of Effective Schooling Approaches and Findings--Macro and Micro Views

#### A Macro View

In reviewing individual studies and syntheses of effective schooling research, we became aware of both macro and micro approaches. Three macro and six micro views are examined in this paper. At the macro level, there is a guiding assumption underlying the research. It is assumed that there are significant differences in levels of educational outcomes across schools that can be accounted for by differences in educational treatment and not just by student socioeconomic status.



Barker ...

-16-

Thus, effective schooling researchers have sought to identify outlier schools (extreme examples on either end of a continuum) whose students score especially high or low on a particular outcome measure such as a studardized test of basic skills. High and low performing schools are then studied to determine what differences in educational practices may account for the differences in measured outcomes.

The application of an outlier design approach to vocational education research could result in a more systematic study of program features that contribute to vocational excellence. In states or districts where competency based measures are used for specific occupational areas or for broader vocational outcomes, the schools or programs scoring especially high or low, after adjusting for pretest scores and the socioeconomic level of the neighboring community, could be selected for a more intensive look. The closer look would focus on specific characteristics of the vocational programs at each end. For example, a state having 50 high school industrial mechanics programs might study what variations in course content and instructional modes result in greater gains of students on competency based performance measures, attitudes toward mechanics, and future placement of students in related positions or in advanced training in this field.

A second feature of many school effectiveness studies is that schools which are found to be high or low on one outcome measure tend to be so on other measures. Thus, Michael Rutter and his colleagues came to the conclusion in <a href="Fifteen Thousand Hours">Fifteen Thousand Hours</a> that "the combined effect was much more powerful than that of any individual factor considered on its own. For this and other reasons, we have suggested that some kind of overall school 'ethos' might be involved" (Rutter 1982). School 'ethos' is described as the social organization of a school with its own rules, values, and standards of behavior.

As applied to vocational education, this could suggest looking at students' attitudes toward being a vocational curriculum or academic curriculum student. Several secondary vocational education studies have already found a much more positive attitude of students toward their vocational program than to their nonvocational program. Participation in vocational clubs may also be a factor in helping students feel a close bonding to their vocational programs.

A third broad feature related to the effective schools studies is the emphasis on collecting and applying the research findings to help districts and schools improve their effectiveness. The national network of educational laboratories and research and development centers has joined forces to disseminate the research findings to schools and to provide technical assistance, whenever possible, in applying the findings to solving school-based problems. In this sense, the effective schooling movement has grown beyond syntheses of research studies to a concentrated attempt to reform educational practice. Vocational education could benefit from participating in these dissemination and school improvement efforts.

#### A Micro View

One reason that effective schooling research has received good press is that it has delivered findings on some very concrete issues facing education today. At the micro level, effective schooling research syntheses have pulled together separate studies to focus on specific issues such as the role of the school principal, staff expectations of students, and teacher behaviors leading to effective classroom learning. In terms of teaching behaviors in instructional programs, one researcher has synthesized much of the recent research around six functions -- review of previous learning, demonstration of new material, guided practice and checking for understanding, feedback and corrections, independent practice, and periodic review (Rosenshire 1982). In this paper we discuss six micro aspects of effective schooling: 1) time on task, 2) expectations of performance, 3) student motivation, 4) cooperative learning, 5) effective learning climate, and 6) use of evaluation. These gix aspects were selected because they were found to be supported across multiple research studies, reflect a full spectrum of findings by researchers from diverse backgrounds, and have strong potential for application to vocational education. It is hoped that other researchers will show application of other effective schooling research findings to vocational education. For example, future attention should be given to the leadership role of the principal and research findings related to teacher effects.

## Limitations of Effective Schooling Research

As in any research synthesis, the conclusions reached about effective schooling at the secondary level are dependent upon the quality and characteristics of the original research studies (Brookover 1981, Purkey and Smith 1982). Unfortunately, many of these studies contain serious limitations. Following are some of the most common limitations:

- The majority of the studies completed were focused on elementary schools. The application of the same findings to America's high schools is questionable.
- o Standardized tests of basic skills are often used as the sole criterion for effectivenenss. However, many of the outcomes considered important for schools 90 beyond basic skills as measured by a particular standardized test.
- One-shot studies have generally been conducted rather than longitudinal studies. Thus, it is not clear whether there would be a recurrence of findings from year to year.
- o Findings across separate studies are often inconsistent. For example, the role that teacher expectations of student performance play in increasing learning is not clear.



-18-

- o In some studies, the effective schools tend to be found more frequently in communities with higher socioeconomic status. Thus, it is not known if the differences are attributable more to differences in student characteristics or to differences in educational treatment.
- Variables often found associated with effective schools may not be the same ones found to change an average or poor performance school into an excellent school. Research on how to improve schools is a separate body of knowledge. Therefore, these factors may not be that helpful in school improvement.

## Linking School Improvement with Effective Schools Research

Research has clearly shown the relationship of certin schooling practices to student achievement. It becomes obvious, rather quickly, when reviewing school effectiveness literature that conclusions regarding specific factors that lead to effective schools are tentative and should be viewed with caution. Variations in findings of school effectiveness studies are rampant; studies of expenditures, facilities and teacher qualifications, for example, have not been consistent in explaining the differences in student achievement as measured by standardized tests. Yet other factors have been shown to be more effective, as evidenced by the preponderance of data generated by the Goal Based Education Program at the NWREL and by other researchers. The work of these groups indicates that research and development processes and results can be used for local school improvement in a variety of contexts. Some practices have been proven to be more helpful than others because they assist educators in providing better instruction; hence, students learn more. The Goal Based Education Program has found that the following factors, when interrelated, are conducive to learning: 1) a clearly defined curriculum, 2) focused classroom instruction and management, 3) ongoing assessment and e-valuation of students and programs, and 4) strong instructional leadership (Effective Schooling Report 1983).

One approach to school improvement looks at the notion of school culture, that is, how at schools link content with process (Rutter 1979, Brookover 1979). Culture refers to the organizational structure: the roles, norms, values, and instructional techniques; to the information imparted in the curriculum (Purkey and Smith 1982). Process includes the political and social relationships of a school as well as to the flow of information. School effectiveness can be determined by a school's climate, composed of attitudes, behaviors, the organizational structure and so on. Therefore, an effective school would be likely to have clear goals related to student achievement, high expectations for students from parents and teachers, and a structure designed to maximize opportunities for students to learn (Purkey and Smith 1982).

The concept of school culture resolves many of the problems associated with school effectiveness studies (Purkey and Smith):



-19-

- O It assumes that changing schools requires changing people—their behaviors and attitudes—as well as school organization and norms.
- O It assumes that consensus among staff is more powerful than overt control, without ignoring the need for strong leadership.
- O It has the advantage of being equally applicable to elementary and secondary schools.
- o It is neither grade level nor curriculum specific.

## IV. FINDINGS FROM EFFECTIVE SCHOOLING STUDIES

#### Design Applications

The National Center for Research in Vocational Education (NCRVE) conducted a comprehensive review of nearly 1500 studies reported since 1968 of the effects of vocational education on participants (Mertens 1980). Their report is based on a synthesis of 88 statewide, regional, or local studies selected on the basis of relevance and meeting minimal standards of methodological acceptability. The NCRVE study found insufficient evidence of vocational education impact in the areas of basic skills attainment, academic abilities, occupational skill attainment, attendance/dropout, attitudes, and values. Procedurally, the studies reviewed were of three types: 1) comparison of vocational education participants on a selected criterion with that of comparison of vocational education participants on a selected criterion with that of comparison/control participants, 2) comparison of vocational education participants' performance with the general population, and 3) simple descriptions of vocational education students, performance on selected criteria. No comparison was found between schools within a district or state demonstrating high performance versus those demonstrating low performance. Nor were there any studies reported that attempted to identify the effects on student performance of program factors such as those studied in effective schooling research.

## Applying Selected Elements of Effective Schooling

Based on our extensive review of effective schooling literature, six factors which have been shown to positively relate to increased learning were selected for consideration by this study: time on task, expectations of performance, student motivation, cooperative learning, effective learning climate, and use of evaluation.

Each of these is supported by empirical data showing positive correlation with learning gains. Additionally, it was felt that these six are those with greatest potential for application to vocational education settings. The general findings of each critical factor are presented followed by suggested applications to vocational education.

## Critical Factor: Time on Task

Researchers have been studying time as an important factor in learning for decades. A direct relationship has been established between achievement and active learning time for both individuals and for groups of students. General findings include:

o There is a clear relationship between active learning time and gains in achievement; the greater the actual time on task, the greater the level of achievement (Bloom 1980).



- The difference in mastery and nonmastery for a student needing corrective instruction is about an hour of extra instruction every two weeks or six minutes of extra time per school day (Bloom 1980).
- O Improving achievement by increasing active learning time requires the cooperative efforts of a principal and staff (Brookover et al. 1982).
- Academic Engaged Time (AET) involves: 1) time planned for instruction, 2) time actually spent on instruction, and 3) time the student is actively engaged in learning the task (Brookover et al. 1982).
- Maximum classroom time should be allocated for instruction (Brookover et al. 1982).
- O The level of school attendance explains a significant proportion of the variance in school achievement; absent students receive less instruction (Wiley and Harneshfeger 1974).
- Academic instructional time differs among various groups within a school; college bound groups receive more hours of academic work than do vocational educational groups or other student groups (Rosenbaum 1976).
- O Teacher reports of time on task are not adequate measures of time usage; only observations by an objective observer give valid measures (Brookover et al. 1982).
- O The Beginning Teacher Evaluation Study (Fisher et al. 1978) suggests that seatwork is one period when students are frequently not on task; absence of direct teacher-student contact during independent work time increases the chance of students not being on task.
- O Transitions and discipline are major detractors from time on task (Brookover et al. 1982).

## Applications of "Time on Task" to Vocational Education Programs

Only one research study was found that applied effective schooling factors to vocational education. Halasz and Behn at NCRVE looked at time on task as an effective schooling factor within the context of vocational education (1983). Previous studies have shown time on task to be one of the most critical factors that affects how much students learn and achieve in school. Empirical data show: 1) time on task positively relates to student achievement, 2) students in academic classes learn more when time for learning is increased and when students have a high rate of success with learning activities, and 3) students achieve more when teachers lecture, demonstrate, and lead discussions. Yet prior to the Halasz/Behn study, time on task had not been applied to the vocational setting; virtually no baseline data exist to show now time is spent in vocational education classes.



The purpose of the Halasz/Behn study was to show how time was spent in ten vocational classes by 186 students and their teachers. The ten classes studied were located in seven comprehensive high schools and area vocational schools at four types of sites-rural, suburban, urban and inner city-in four states. Three program areas were represented: agricultural education; marketing and distributive education; trade and industrial education. Two project staff spent two weeks in each class observing and recording students' and teachers' activities each minute with observation instruments designed for the study. The instruments were designed to show the amount of time students spent on task upon content (basic skills, technical skills, esployability skills) as compared with time spent on noncontent (set up, clean up, etc.) and off task (time spent on breaks, socializing, and so on). Teacher time was recorded either as content time or as managerial, noncontent time.

Implications of the study as derived by Halasz and Behn include:

- 1. The average proportion of time on task in vocational programs is very similar to the amount of time on task found in studies conducted in academic areas.
- 2. A wide variation of time on task was found among the ten classes, especially on content. This indicates that many factors contribute to time usage in secondary vocational education.
- 3. Students' observed time on task (55.9 percent) was consistently less than time allocated by teachers (67 percent).
- 4. Longer classes and classes with fewer students had the highest proportion of time on task.
- 5. Classes with more hands-on activities (trade and industrial and agricultural education) had more time on task than lectureoriented classes (marketing and distributive education).
- 6. Results of the study show vocational education classes are diverse; there is no "typical" class that could be representative of all vocational education classes.

Halasz and Behn recommend that since no attempt was made to relate outcomes to time on task, further studies need to investigate this relationship. They feel additional research is needed to examine the relationship of teachers managerial activities and instructional methods to time on task.

In reviewing this study, three principles seem to emerge when applying effective achooling concepts to vocational education. First, the differences in learning environment must be recognized. Prior time on task measures and instruments were designed largely for single group/total class activity generally led by a teacher. In contrast, the NCRVE team had to develop new categories appropriate to a hands-on



-23-

setting in which individuals or small groups of students may be working separately. Second, the proportion of time on task from nonvocational classes provides an interesting base for comparison with vocational education as long as one also accounts for the different purposes and activities in these two settings. Third, no "typical" vocational class was found. Thus, it is important to compare findings among areas of vocational training.

## Critical Pactor: Performance Expectations

Researchers have long noted the link between teachers' expectations and student achievement. What teachers expect students to learn is likely to result in what students will learn. General findings include:

- Teacher expectations are a strong influence in setting learning climate; the processes by which expectations are formed and transmitted to pupils are largely unconscious, becoming a part of the school learning climate (Brookover 1982).
- elements of effective schools include: 1) administrators and teachers have high expectations for student achievement, 2) expectations are clearly defined, 3) general expectations are not reduced for any student, 4) students are expected to work toward priority goals, and 5) everyone accepts the notion that school is a place for learning (GBE Program document 1983).
- o When students are not successful in academic areas or when academic work is unrelated to current and future goals, they look for recognition outside the school setting; peer social judgment becomes more important as academic success declines (Faunce 1979).
- O High expectations result in increased and more effective instruction; low expectations result in decreased and less effective instruction (Brophy and Good 1974, Finn 1972, Persell 1977).
- O Teachers' expectations have greater influence on low income and/or minority students than on middle class students (Brookover et al. 1979).
- Parents' expectations and evaluations influence their children (Brookover 1982).
- Almost all students are capable of reaching competence on age/grade level objectives (Bloom 1976).
- O Students conform to the level of expectations teachers set (Crano and Mellon 1978, Rist 1970).



-24-

- Student expectations are affected by teachers' predispositions in four major areas (Persell 1977):
  - Teachers' personality traits

the property of the property of the control of the property of the control of the

- 2. Teachers social experiences and prejudices within the school as well as within the community
- 3. Teachers' concepts and beliefs regarding IQ or cultural patterns:
- 4. Teachers knowledge and use of such educational structures as grouping arrangements of testing programs

Lowered academic expectations of performance are associated with the following (Brookover 1982):

- Sex-lower expectations are often held for older girls due to sex role discrimination.
- o Social Economic Status (SES) -- lower expectations are often held for lower SES (including level of parental education, types of jobs held, place of residence, and so on).
- Race--lower expectations are often held for minority status students.
- o Test scores, permanent records—a belief in the "fixed ability" of students often precludes the possibility of improvement and higher expectations.
- Negative comments about students—lounge talk and other teachers' or principal's negative evaluations can result in lower expectations.
- Type of school—rural and inner city schools are often associated with lower expectations.
- Appearance—-lower expectations are often associated with type, style and expense of clothes, and grooming habits.
- Oral language patterns—negative cues from any nonstandard English speaking stunt often result in lower expectations.
- o Tracking or grouping—labeling effects and a tendency to accentuate differences among students often results in lower expectations.

## Applications of "Performance Expectations" to Vocational Education

Performance expectations in vocational education are sometimes introduced through the use of standardized achievement tests in various occupational fields. For example, the National Occupational Competency Testing Institute (NOCTI) has been working with forty-seven states and the District of Columbia to develop both teacher and student occupational



competency examinations (NOCTI 1983). NOCTI has developed Student Occupational Competency Achievement Tests in 19 occupational areas and is currently field testing instruments in 11 additional areas. These written and performance tests reflect the expectations of both vocational teachers and people working in the occupational areas. The results are being used by vocational administrators to assess strengths and weaknesses of current programs and present curricula. Some employers are using the examination for hiring individuals on the basis of demonstrated levels of competence as opposed to letter grades.

Other ways that findings on performance expectations could be applied include:

- o Performance expectations in vocational classes need to be reviewed by vocational coordinators, teachers, and vocational advisory committees to determine their appropriateness in relation to students, abilities, available time, and the changing needs of the workplace. Local advisory committees are useful in helping to identify employer expectations for various occupations.
- Teachers' performance expectations for vocational education students need to be clearly communicated to the stude is and parents. They need to be monitored throughout the course so that students are continuously aware of what is expected and how their individual performance matches up with these expectations.

#### Critical Factor: Student Motivation

An effective school learning climate includes high expectations for all students to succeed. Researchers have found various reinforcement techniques will help motivate and sustain appropriate student behaviors. For example, Hunter found six teacher behaviors have a powerful effect on student motivation (1979):

- Concern—students have optimal levels of concern that motivate them to greater effort and higher performance; teachers must sometimes raise, sometimes lower, levels of concern.
- 2. Feeling tone--whether pleasant, unpleasant or neutral, use of tone depends on the situation, but teachers must return to pleasant tone as soon as the student has performed.
- 3. Interest--students' interest can be developed by 1) making the learning more meaningful and 2) making the learning more vivid or different from what the student usually experiences or is expecting.
- Success—motivation increases as success increases; however, some effort is required of the student to feel successful.



のないないのではないのとなるないないないないというないというないないないないないないないない

-26-

Andrew Edition of the Constant of the Constant

- 5. Knowledge of results—students are highly motivated if they know how they're doing while they're learning.
- 6. Intrinsic vs. extrinsic motivation—intrinsic motivation is developed when an activity or the learning itself is the reward for effort; extrinsic motivaton is developed when the student learns to gain status, approval, grades, or other rewards.

Other general findings regarding student motivation as it relates to learning gains include:

- O Positive reinforcement (rewards) are linked to motivation; when the behavior (whether right or wrong) is positively reinforced, the probability of that behavior being repeated is high (Brookover et al. 1982).
- Negative reinforcement used in a nondiscriminating manner is likely to impede student learning (Brookover et al. 1982).
- o Praise is relatively easy to apply and is a direct statement of the contingency between the behavior and the reinforcer (Brophy 1981).
- When intrinsic motivation is absent, extrinsic rewards are necessary (Brophy 1981), but intrinsic motivation is a more powerful motivator and more enduring (Crandall, Good and Crandall 1964).
- Teacher praise is not a useful reinforcer in the classroom because it's often misused (Brophy 1981). Most Praise is noncontingent on students' behavior (Harris and Kapche 1978). Inappropriate praise is found among teachers who have low expectations for student learning (Weinstein 1976, Brookover et al. 1978).
- o For praise to motivate behavior effectively it must be contingent on performance, specific to the behavior being reinforced, and be sincere and convincing (O'Leary and O'Leary 1977).

## Applications of "Student Motivation" to Vocational Education

Several vocational education studies have looked at the reasons why students enter a high school vocational education program. For example, in a statewide study of secondary vocational education, it was found that the most common reasons students gave for entering vocational education programs were an interest in the content area and to learn specific job skills (Owens 1982). Students entering vocational education to learn specific job skills or who had an interest in the area were significantly more likely to report planning to get a job or taking further training in the occupational area than those who entered because their friends were enrolled or because of advice from parents or others.



-27-

Vocational clubs such as Future Business Leaders of America (FBLA) and Future Farmers of America (FFA) provide both intrinsic and extrinsic motivation to learn. Public recognition for projects well done is often a strong positive reinforcement for young people. Thus, in looking at excellence in vocational education it becomes important to look not only at what is occurring in the classroom, shop, or laboratory, but also at the informal learning taking place in the community.

Vocational education can also motivate certain students to learn through hands-on experiences who would otherwise be turned off to learning through books and teacher lectures. In a national survey, Donna Mertens and her colleagues (1983) found:

Students classified as potential dropouts were more likely to finish the tenth grade if they participated in vocational programs. When they reached the legal age of withdrawal from school the following year, however, members of this group tended to leave regardless of the educational program in which they were enrolled. Once past this critical point, vocational education was able to hold the remaining high-risk students in school through graduation at a higher rate than other education programs (AVA Fact Sheet 1983).

In a statewide study of vocational education in Oregon (Owens 1982), students, parents, teachers, and advisory committee members were asked to rate student outcomes resulting from participation in vocational education programs. "Willingness to learn new things" was rated particularly high by each of these groups, indicating an important motivational outcome of vocational education.

## Critical Factor: Cooperative Learning

In addition to general findings on student motivation, there has been extensive research on cooperative (team) learning as a motivator. General findings about cooperative learning include:

- o Groups have powerful effects on student behavior and motivation; groups reinforce attitudes, actions, and behaviors or the members (Slavin and DeVries 1979).
- Peer relationships strongly influence student feelings about the importance of behaviors, skills and attitudes (McDill and Rigsby 1973).
- Research on cooperative learning techniques prove them to be effective in increasing student performance and positive self-concepts (Slavin and DeVries 1979, 1980; Aronson et al. 1975).



-28-

, 6

- o For group and team competition to work, teahers must effectively use tangible and symbolic rewards that are attractive to the student peer culture (Brookover et al. 1982).
- Team learning research suggests that students prefer to and learn better when they teach and learn from each other (Devin, Sheehan et al. 1976).
- o The addition of an effective team incentive system to mastery learning techniques results in greater achievement than either component by itself (Slavin 1983).
- O Cooperative learning methods result in positive: 1) race relations, 2) attitudes toward academically handicapped students, 3) self-esteem, and 4) ability to cooperate in other settings (Slavin 1983).
- Among cooperative learning methods where students study the same material together, only those that provide group rewards based on individual learning consistently increase student achievement; group rewards and individual accountability are held to be essential to the instructional effectiveness of cooperative learning methods (Slavin, 1983).
- Cooperative learning methods include cooperative incentive structure and cooperative task structure; the critical feature is that two or more students are interdependent for a reward they will share if they are successful as a group (Slavin 1983). Cooperative incentive structure involves rewards for both group and individual learning. Cooperative task structure involves two or more students allowed, encouraged, or required to work together on a task, coordinating their efforts.
- Two factors must be present if cooperative learning methods are to be more instructionally effective than traditional methods:

  1) group reward and 2) individual accountability (Slavin, 1983).

## Application of "Cooperative Learning" to Vocational Education

Vocational education programs provide excellent settings for cooperative learning and team competition. The latter competitions can occur especially through vocational clubs that may involve group projects. Research on cooperative learning in effective schools suggests that vocational teachers should: 1) disucss with students the concept of team learning; 2) form teams that reflect a range of performance, sex, and ethnic mix of the class, 3) select rewards that are attractive to the student peer culture, and 4) observe both individual and group accountability.

Vocational teachers typically use a variety of group activities including:

Livestock judging teams at county and state fairs

ाक दूरा द्वार क्राप्ति हर द्वार व

- Model or simulated office operations
- O Auto diagnosis and repair contests
- O School landscaping project

- Construction of an actual home
- o Planning and implementing a fashion show

Additionally, school-endorsed programs such as Junior Achievement, Explorer Scouting, and 4-H accomplish their activities through cooperative learning activities. The National Commission on Resources for Youth has been a strong advocate for cooperative learning and has developed useful guidelines for what they call a collaborative group (1983). They particularly emphasize the need fo structure, flexibility, and responsible roles for youth.

## Critical Factor: Effective Learning Climate

Researchers have found schools that have effective learning climates also have high levels of achievement; conversely, schools with ineffective learning climates produce low levels of student achievement. General findings include:

- Effective school learning climate is comprised of "appropriate conditions for learning." It refers to attitudinal and behavioral patterns in a school which affect levels of achievement (Brookover 1982).
- School learning climate is also defined as "the norms, beliefs and attitudes reflected in institutional patterns and behavior practices that enhance or impede student learning" (Lezotte et al. 1980).
- Three clusters of school characteristics make a difference in the level of achievement and have an interactive relationship: ideology, organizational structure and instructional practices. There is a basic and important distinction between school climate and school learning climate (Brookover 1982):
  - 1. School learning climate relates to student achievement and those factors within a school that affect achievement.



4

William Barbara Control of the Sec.

30- 3

2. A school's learning climate is the collective set of attitudes, beliefs, and behaviors within a building. It goes beyond the individual to the group norms of a school. These norms tend to be maintained over time with new members being socialized into the prevailing set of behaviors.

and the contract of the way of the same

- 3. The school learning climate describes the school as a social system. Since schools share a common function in society, there is some similarity in learning climates. On the other hand, different schools stress different "philosophies," instructional practices and methodologies, and beliefs and expectations of students' abilities to learn. Consequently, school learning climate varies sufficiently to produce different levels of student achievement.
- 4. The school learning climate can be changed. Local and building norms do change and the people who are members of the school social group are the change agents. Outsiders are unlikely to have much aspect on the social group unless that group desires or is willing to change.
- School learning climate can explain as much of the achievement differences among schools as do SES and race (Lezotte et al. 1980).
- School learning climate as an effect on student achievement is documented in other countries: Israel junior high schools (Chen and Fresko 1978); high schools in Ireland (Madaus et al. 1979); London, England high schools (Rutter et al. 1979). The results of the foreign studies support the work done in high schools in this country (Brookover et al. 1979; McDill, Meyers, and Rigsby 1967; Glasheen, Hadley, and Schneider 1977).

## Applications of "Effective Learning Climate" to Vocational Education

In applying research on "effective learning climate: to vocational education it is useful to first distinguish among three learning environments—the total high school, the vocational program as a subculture within the school, and the workplace. Students often have quite different feelings regarding these three environments and the learning culture of each setting. Some vocational students we have interviewed feel a more personalized attachment to the vocational program or work site setting than they do to the rest of the school. They see their role and the expectations of adults toward them as entirely different from one environment to another.

In a study of the workplace as a learning environment, 1100 senior high school students who were enrolled in Experience-Based Career Education Programs in 16 states were surveyed (Owens and Owen 1981). The research centered on the job site characteristics youth associate with excellent or with poor community learning experiences. Factors found to



to act of the section of the section of the section of

lead to excellent learning at job sites included the opportunity for hands-on learning and for applying the learning to new things, positive interactions with adults, being given adult responsibilities, having clear directions to follow, and being given challenging tasks.

A study focusing on developing responsibility in youth centered on the role played by the home, the school, and the workplace (Owens 1983). In the perception of the young people interviewed, high school helped them become more responsible by being expected to be on time and to do nomework. However, many students felt they were not treated as adults in most classes. Their vocational and career education programs were often perceived as a more trusting and supportive environment. Being at a work site was seen by students as an opportunity to work independently, be trusted as an adult, and to feel "worth something."

Vocational educators need to consider the ways in which they can improve the learning climate in each of the three settings. At the total school level, vocational teachers can cooperate with the principal and other staff by establishing and following consistent procedures regarding student attendance, grading, and other common functions. Within the vocational education department, they might agree to enforce some additional common standards in areas such as snop safety. For students involved in cooperative education or other work-related programs, the vocational staff can help to establish an effective learning climate by working closely with employers in agreeing on student placement objectives, monitoring student performance at these sites, and providing positive reinforcement to students for excellent work performed at these sites.

## Critical Factor: Use of Evaluation

An integral part of developing and maintaining an effective school learning climate is regular and consistent use of evaluation techniques. The efficient use of assessment data is an important tool in making decisions regarding program improvement. General findings include:

- Teachers and other staff can increase the level of student achievement in their classroom when they use assessment data to guide needed curricular and instructional modifications (Brookover et al. 1982).
- Objective-referenced teacher-made tests allow for more frequent modification of instruction by furnishing measurements of classroom achievement of specific objectives over shorter periods of time (Block and Anderson 1975).
- Oral testing should be used with students who have reading problems or react negatively to written tests (Brookover et al. 1982)...



-32-

- o Frequent progress tests serve two purposes: 1) they familiarize students with testing through practice and 2) they provide assessment information more frequently (Brookover et al. 1982).
- Norm-referenced tests are a poor measure for many instructional programs; tests must focus on the same content as is actually taught in the classroom (IRT study 1979).
- o In effective schools:
  - 1. Staff know how students are doing and let students know.
  - 2. Student assessment is planned, regular, and routine and causes minimal disruption of classroom instruction.
  - 3. Testing results are reported quickly in terms of learning objectives on which students were tested.
  - 4. at the district level, clearly stated assessment procedures are carried out by district staff; major tests are announced well in advance to all time for building and classroom scheduling; there are specific routines for scoring, storing, reporting, and analyzing results.
  - 5. at the building level, staff follow simple routine procedures for frequent collection, summarization, and reporting of student achievement information; individual student records are maintained regarding achievement, discipline, and tardiness; and
  - 6. at the classroom level, teachers frequently monitor student learning, both formally and informally; teachers know and use test development techniques to design valid, reliable assessment instruments; assessment is routinely handle and students quickly hear results; reports of results to students are simple, clear and tied to learning objectives (GBE Program Document 1982).

## Applying "Use of Evaluation" to Vocational Education

Evaluation is used in vocational education to: 1) assess teachers' occupational competency, 2) assess student readiness for particular learning, 3) evaluate student progress and mastery of learning objectives, and 4) evaluate program effectiveness. As in effective schooling literature, it is important to plan not only how the data will be collected but also how they will be reported and used for student and program improvement.

Unlike academic teachers who rely heavily on standardized and teacher-made tests, vocational educators have a greater opportunity to use competency based performance measures, computer simulations, direct observation, and product evaluation based on standards found in business and industry. Performance tests can be used to provide students with



explicit feedback on which skills they have mastered and which ones require further practice. Direct observation of student behaviors is especially effective for judging students' understanding of safety precautions and correct assembly or performance procedures. Because vocational classes often result in students producing tangible products, teachers and students can evaluate the physical products using commonly accepted industrial standards. Thus, there is often less subjectivity in these evaluations than might be true in grading student essays.

2-45.E

An evaluation issue being faced by some school districts is the attempt to agree upon and use common performance tests across all classes with the same title. This approach has the advantage of assuring employers that students who have passed a particular automotive course, for example, will also be able to demonstrate a minimum set of skills regardless of the school attended.

A standard approach to this problem is a modified job or task analysis procedure whereby local employees validate a set of employability skills or occupational tasks. Some districts then utilize a skill development record or card as a way to document competence.

This section of the paper has attempted to describe selected principles of effective achooling research and show how they are or could be applied to secondary vocational education. The following section will use a similar approach based on principles learned from successful businesses.

### V. FINDINGS PROM STUDIES OF SUCCESSFUL BUSINESSES

### The Peters-Waterman Study

Supporting future directions for research to improve vocational education are principles of excellence identified by Thomas J. Peters and Robert H. Waterman, Jr. We believe principles of excellence resulting from Peters; and Waterman's study of 62 of America's best-run companies provide some useful insights into achieving excellence in school career and vocational programs as well. First, however, cautions are in order. The bottom line profit motive of the private sector is not applicable to public education. Nor are achools free to determine what to produce or who their clients will be: Some of the eight principles that Peters and Waterman derived from studying large corporations may not apply to all vocational settings. Just as they may not apply to smaller school districts or smaller businesses. Likewise, certain principles may be more relevant to vocational education than others. Indeed, Peters himself in a recent interview with U.S. News and World Report, expressed the belief that a bias for action, closeness to the customer, and high productivity through people (see definitions that follow) are the three most important qualities for coporate success.

The principles developed by Peters and Waterman warrant serious consideration and suggest some exciting implications for secondary vocational education, particularly when we know that graduates will soon be entering the workplace where these principles may be applied.

### Description and Application of Principles

In Search of Excellence delineates eight principles: 1) A Bias for Action; 2) Close to the Customer; 3) Autonomy and Entrepreneurship; 4) Productivity Through People; 5) Hands-On, Value-Driven; 6) Stick to the Knitting; 7) Simple Form, Lean Staff; and 8) Simultaneous Loose-Tight Properties. A description of each principle is given followed by applications to the vocational education setting.

A Bias for Action. The more successful companies decide on new directions very quickly without waiting for extensive analysis of issues before moving. One business establishment uses the motto, "ready, fire, aim"—a maxim implying their willingness to take action and then refine actions as they proceed.

Close to the Customer. Listening to customers' advice and responding to changing customer requests is needed for success in business. The importance of the customer is kept in mind, not just costs and profits.

Autonomy and Entrepreneurship. Personal leadership and self-initiative are recognized and rewarded in successful companies. Certain people are valued who take responsibility for moving products from the invention to the marketing stage.



-35-

Productivity Through People. Peters and Waterman suggest increased productivity can occur by rewarding successful employees and treating them as adults. Quality circles are one example of how employee ideas are taken seriously and used.

mail marker in an erhand or have a constant

Hands-On, Value-Driven. Successful executives select a few key values and stress those values in all aspects of business. For example, "teamwork" is not only an advertising slogan for Delta Airlines, but is emphasized in the company's policies and day-to-day operations.

Stick to the Knitting. Successful companies, according to Peters and Waterman, concentrate on those things they are experienced with and do well. They are reluctant to enter unrelated fields in which they have no experience or knowledge.

Simple Form, Lean Staff. Large corporations in the 1980s are finding ways to keep the number of central office staff to a minimum. Newer management styles of some high technology firms are calling for less hierarchy, greater staff accessibility to executives, and less need for middle management personnel.

Simultaneous Loose-Tight Properties. Becouraging decision making at the local level while retaining central monitoring of a few key factors is the final success principle discussed by Peters and Waterman. An example of this principle in business is IBM top management knowing within hours when the company has lost a corporate customer. While such information is gathered companywide, other IBM functions are highly decentralized.

The following table displays each principle of excellence, giving an example of what is currently being done in some vocational education program or what could be done. This in no way is intended as a comprehensive application of these principles, but merely as an illustration of possibilities. School and district staff could use these principles to generate a more complete model.



### TABLE 5

### APPLICATION OF EXCELLENT BUSINESS PRACTICES TO VOCATIONAL EDUCATION

### What is being done

# A Bias for

Many vocational educators are skilled in moving quickly from subject matter theory to involving students in hands-on applications. Diversified occupations classes and cooperative education provide immediate opportunities for learning in new and emerging vocations.

### Close to the Customer

Action

Imput from local advisory committees for some occupational clusters is helping vocational instructors keep their courses up-to-date. Student and employer follow-up studies in some districts are used to gather "customer" feedback that is used for program improvement

## Entrepreneurship

Autonomy and Some vocational programs give students excellent experiences in forming and operating "companies" to produce and market products and services. Junior Achievement is one example of an out-of-school program with similar objectives.

### What could be done

As local labor market conditions change, high school vocational educators need to become adept in dropping programs showing no demand for employment and in adding training for high demand occupations. Short term training-often felt to be a postsecondary responsibility-may need to be considered in high school as well. A short course in running electronic cash registers or waiting tables may give students higher preference in a competitive job market.

With the increasing need for economic development collaboration, schools and community colleges need to work together in anticipating local employers; specific needs. Encouraging employees to teach segments of vocational courses in which they have special expertise would also help ensure that students are receiving up-todate instruction in areas of importance to local employers.

Employers are calling on schools to do a better job in helping students understand the free enterprise system. Some corporate executives might be excellent classroom speakers to address this topic and arouse student interest.

### What is being done (cont.)

What could be done (cont.)

Productivity through People

Students in cooperative education programs are often treated like regular employees while in the workplace and their suggestions are sometimes used to improve operations

In the vocational classroom and shops, students should be treated as responsible adults and rewarded for excellence in individual and group performance

Hands-On,

Students in effective Value Driven vocational programs leave with mastery of transferable occupational skills and good work habits that have been reinforced in vocational courses as well as through the social climate of the entire school.

At the school and district level, confusion often exists as to primary purposes of vocational education. Clarification and agreement should be reached as to whether particular courses are designed primarily to orient students to the world of work and/or to teach essential occupational skills needed for job entry.

Stick to the Knitting

Some schools are successfully emphasizing applied basic skill:, good work habits, and essential occupational skills in vocational areas.

High schools may need to reassess whether, in attempting to offer a wide variety of occupational areas and specialty courses, they may be reducing the quality of some basic vocational offerings.

Simple Form, Lean Staff

Budget cuts have forced reductions in the number of vocational specialists at the state and district level. This can sometimes force the central staff to take a more integrated approach to vocational education.

Improved collaboration with community colleges and private industry may allow some students to receive credit for specialized training taken outside of high school and may also result in climinating duplication of services.

Loose-Tight Properties

Simultaneous The entire staff in some schools has agreed on common guidelines for handling student discipline problems while allowing individual teachers freedom to work within those guidelines.

Districts could establish districtwide minimum competencies required for students to complete a particular occupational class. Additionally, they could give maximum freedom to teachers helping students master these competencies and in deciding optional competencies to add.

### IV. FINDINGS FROM SCHOOL IMPROVEMENT POLICY STUDIES

Larry Cuban has drawn attention to the fact that in the school effectiveness movement, little attention has been given to the role of district leadership (Cuban, 1984). He describes the role of the school board and superintendent in establishing districtwide instructional goals, revising student procedion and graduation policies, mandating planning processes, reviewing the curriculum, revising district supervisory and evaluation practices, creating a districtwide assessment program, and introducing staff development programs. Cuban goes on to suggest mixing of top-down and bettom-up change strategies.

Several empirical studies have been done recently that examine the relationship of school district policies to secondary school improvement. In the Cosmos Corporation study of excellence in urban high schools, the authors point out the in-dequacies of theories that view the high school as an independent autonacus unit as well as those that view change only from a district perspective (Yin, Blank and White, 1984). Instead, they propose a third perspective in which the school and district "co-manage" the school. This perspective became the base for the research design in their District/Secondary School Study.

The "co-management" perspective was also used in the secondary school improvement study conducted by the Northwest Regional Educational Laboratory for the Council of the Great City Schools (Owens and Hathaway, 1986). This led to a survey not only of superintendents of the 35 largest school districts, but also of a sample of high school principals within these districts. The findings covered the policies that relate to secondary school improvement, how these policies were being carried out, preliminary impact notices, and perceived district support most helpful in contributing to secondary school improvement.

As applied to vocational education, the school improvement policy literature would suggest the importance of a study examining policies designed to improve vocational education effectiveness and the interaction of the district and school in bringing about improvement in vocational education.



٤.

-39-

### **BIBLIOGRAPHY**

- Aronson, E. et al. The Jiqsaw Classroom. Beverly Hills, CA: Sage Publications, Inc., 1978.
- Behr, G. "Test-Wiseness: Using Test Information for Planning Instruction." Paper presented at the Alaska Regional Conference. Northewest Regional Educational Laboratory, 1982.
- Block, J. and L. Anderson. Mastery Learning in Classroom Instruction. New York: McMillan Publishing Co., 1975.
- Bloom, B.S. "Implications of the IEA Studies for Curriculum and Instruction." In A.C. Purves and D.U. Levine (eds.), <u>Educational Policy and Inernational Assessment</u>. <u>Implications of the IEA Surveys of Achievement</u>. Berkleley, California: McCutchan, 1975.
- Bloom, B.S. <u>Human Characteristics and School Learning</u>. New York: McGraw-Hill, 1976.
- Bloom, R.B. "Teachers and Students in Conflict: The CREED Approach."

  Phi Delta Kappan. 1980, v6ln9, pp. 624-626.
- Bottoms, G. "Voc Ed Belongs on the National Agenda for Excellence." Vocational Education. v58, 1983, pp. 8-9.
- Bowles, S. and H.M. Levin. "The Determinants of Scholastic Achievement: An Appraisal of Some Recent Evidence." <u>Journal of Human Resources</u>. 1968.
- Boyer, B.L. <u>High School: A Report on American Secondary Education</u>. The Carnegie Foundation for the Advancement of Teaching, No. 5 Lyy Lane, Princeton, N.J., 1983, 355 pp.
- Brookover, W.B. Effective Secondary Schools. Philadelphia: Research for Better Schools, Inc., 1981.
- Brookover, W.B. et al. <u>Creating Effective Schools</u>. <u>An Inservice Program for Enhancing School Learning Climate and Achievement</u>. Holmes Beach, Florida: Learning Publications, Inc., 1982.
- Brophy, J.E. "Teacher Praise: A functional Analysis." Review of Educational Research. 1981, v5ln1, pp. 5-32.
- Brophy, J.E. and T.L. Good. <u>Teacher-Student Relationships: Causes and Consequences</u>. New York: Holt, Rinehart and Winston, 1974.
- Coleman, J.S. The Adolescent Society. The Social Life of the Teenacer and its Impact on Education. New York: Free Press, 1961.
- Coleman, J.S. et al. <u>Equality of Educational Opportunity</u>. Washington, D.C.: Office of Education, U.S. Department of Health, Education and Welfare, 1966.



and the state of whater is in the state of

2111

Copa, G.H. <u>Potential Vocational Education Indicators: Vital Statistics</u> for Planning, Review and Public Information About Vocational Education. Minneapolis: Minnesota Research and Development Center for Vocational Education. June, 1982.

Corcoran, T.B. and B.J. Hansen, The Quest for Excellence: Making Public Schools More Effective. Trenton, New Jersey: New Jersey School Boards Association, 1983.

Crano, W.D. and P.M. Mellon. "Causal Influence of Teachers' Expectations on Children's Academic Performance: A Cross-Logged Panel Analysis."

Journal of Educational Psychology. 1978, v70nl, pp. 39-49.

Crohn, L. <u>Technological Literacy in the Workplace: A Policy Study</u>. Northwest Regional Educational Laboratory. September, 1983.

Cuban, L. (1984, May) Transforming the Frog into a prince: Effective Schools Research and Policy at the District Level. <u>Harvard Education</u> Review, 54, (2), 129-51.

Darcy, R.L. <u>Vocational Education Outcomes:</u> Perspective for Evaluation. National Center for Research on Vocational Education. n163, 1979.

Devin-Sheehan, L. et al. "Research on Children Tutoring Children: A Critical Review." Review of Educational Research. 1976, v46n3, pp. 335-385.

Dillman, D. (1978) <u>Mail and Telephone Surveys: The Total Design</u> Method. New York: Wiley-Interscience Publication.

Edmonds, R. (1984, Fall) School Effects and Teacher Effects. Social Policy, 15, (2), 37-39.

"Effective Schooling Practices." Alaska Research on School Improvement Draft. Goal Based Education Program. Northwest Regional Educational Laboratory. 1983.

Emmer, B.T., Evertson, C.M., Sanford, J.P., Clements, B.S., and Worsham, M.E. (1984) Classroom Management for Secondary Teachers. Englewood Cliffs, N.J.: Prentice-Hall, Inc.

Emmer, B.T. and Evertson, C.M. (1981) "Synthesis of Research on Classroom Management. <u>Educational Leadership</u>. 38, 342-347.

Parley, J. <u>Vocational Education Outcomes: A Thesaurus of Outcome</u>
Questions. Columbus, Ohio: The National Center for Research in
Vocational Education, 1979.

Faunce, W.A. "School Achievement, Social Status, and Self-Esteem."

Paper presented at American Sociological Association Annual Meeting.

Boston, Mass., August, 1979. .

Finn, J.D. "Expectations and the Educational Environment." Review of Educational Research. 1972, v42n3, pp. 387-410.

Fisher, C.W. et al. <u>Teaching and Learning in the Elementary School: A Summary of the Beginning Teacher Evaluation Study</u>. San Francisco: Technical Report VII-1, Far West Laboratory for Educational Research and Development, 1978. (ERIC Document Reproduction Services No. ED165 322)

Fitzpatrick, K. (1981) <u>Successful Management Strategies for the Secondary Classroom</u>. Unpublished manuscript.

Getzels, J.W. "A Social Psychology of Education." In G. Lindzey and E. Aronson (eds.), The Handbook of Social Psychology (2nd ed.). Volume 5, Reading, Mass.: Addison-Wesley, 1969.

Glasheen, J.D. et al. "Student Adaptation of High School Social Groupings and Normative Environments." Paper presented to American Educational Research Association. April, 1977, (ERIC Document Reproduction Service No. ED:143 929).

Goodman, S.M. The Assessment of School Quality. Albany, N.U.: The State Education Department of New York, 1959.

Halasz, I.M. and Karen S. Behn, <u>Time on Task in Selected Vocational</u>
<u>Education Classes</u>. Columbus, Ohio: The National Center for Research in Vocational Education. 1983.

Harris, A. and R. Kapche. "Problems of Quality Control in the Development and Use of Behavior Change Tachniques in Public School Settings." Education and Treatment of Children. 1978, vl, pp. 43-51.

Hedin, D. "Youth Participation in Youth Smployment Programs."

Preconference Papers on the National Youth Participation Conference. New York: National Commission on Resources for Youth, Inc., 1980, pp. 1-8.

Hunter, M. Retention Theory for Teachers. Bl Segundo, CA: TIP Publications, 1979.

Kleinfield, J. and A. Shinwin. "Lessons Out-of-School: Boy Scouts, Girl Scouts and 4-H Clubs as Education Environments." Paper presented at the American Educational Research Association, Montreal, Canada, April, 1983.

Koehler, V. "The State of Teaching: A Summary of Recent Research Findings." Education Times. v14, n3, April 28, 1982, p. 7.

Lathrop, R.L. "The American Educational Structure." <u>Vocational Guidance</u> and Human Development. Edwin L. Herr (ed.). Boston: Houghton Mifflin, 1974.

Lavin, D.E. The Prediction of Academic Performance. New York: Russell Sage Foundation, 1965.

Levin, H. "Vocational Education or College Prep? Opening the Door to Work." IFC Policy Notes. Institute for Research on Educational Finance and Governance. Stanford University, v4, 1983.



.. '.. ' .. '.

それない はんきゃ

Lezote, L.W. et al. School Learning Climate and Student Achievement: A Social Systems Approach to Increased Student Learning. Tallahassee, Florida: National Teacher Corps, The Florida State University Foundation, 1980.

Madaus, G.F. et al. "The Sensitivity of Measures of School Effectiveness." <u>Harvard Educational Review</u>. 1979, v49n2, pp. 207-230.

Madaus, G. et al. <u>School Effectiveness: A Reassessment of the Evidence</u>. New York: McGraw-Hill, 1980.

Mann, D. Chasing the American Dream. Jobs, Schools, and Employment Training Programs in New York State. New York: Community Service Center, 1980.

McDill, B.L. and L.C. Rigsby. <u>Structure and Process in Secondary Schools: The Impact of Educational Climates</u>. Baltimore: The Johns Hopkins University Press, 1973.

McDill, B.L. et al. "Institutional Effects on the Academic Behavior of High School Students." Sociology of Education. 1967, v40, pp. 181-189.

Mertens, D., D. McElwain, G. Garcia and M. Whitmore. <u>Effects of Vocational Education on Participants</u>. Columbus, Ohio: The National Center for Research in Vocational Education, May, 1980.

Mertens, D., P. Seitz, and S. Cox. <u>Vocational Education and the High School Dropout</u>. Columbus, Ohio: The National Center for Research in Vocational Education. Ohio State University, 1983.

Mollenkopf, W.G. and D.S. McCircle. <u>A Study of Secondary School</u>
Characteristics as Related to Test Scores. Research Bulletin 56-6.
Princeton, N.J.: Educational Testing Service, 1956.

National Commission on Excellence in Education. A Nation At Risk: The Imperative for Educational Reform. A Report to the Nation and the Secretary of Education, Washington, D.C., April 1983, 65 pp.

National Commission on Resources for Youth. "Toward Excellence: Developing High Standards in Youth Programs.: Boston: NCRY, 1983.

National Occupational Competency Testing Institute. "Student Occupational Competency Achievement Testing." Big Rapids, MI: Ferris State College, 1983.

National Research Council. Education for Tomorrow's Jobs. (S. Sherman, editor). Washington, D.C.: National Academy Press, 1983.

Northwest Regional Educational Laboratory (Goal Based Education Program) (1984) Effective Schooling Practices: A Research Synthesis. Portland, Oregon.

McClure, L. and Owens, T. <u>Prospectus: An investor's guide to career and vocational/technical education in the Portland Public Schools</u>, Portland, Oregon: Portland Public School District, 1986.



minister of the tenterialists which with the tenter

1 Se al a Backer

دا مارين

- O'Leary, R. and S. O'Leary (eds.). Classroom Management: The Successful Use of Behavior Modification. (2nd edition). New York: Pergamon, 1977.
- Owens, T.R. "Private Sector Views of Vocational Education: A Statement Employer Survey." Paper presented at the annual meeting of the American Vocational Association in Anaheim, CA, December, 1983.
- Owens, T. "Helping Youth Become More Responsible." Paper presented at the Annual Meeting of the American Educational Research Association in Montreal. April, 1983.
- Owens, T.R. Implementation of the Oregon Vocational Education Secondary Effectiveness Study. Prepared for the Oregon Department of Education, Division of Vocational Education. Northwest Regional Educational Laboratroy, 1982.
- Pascarelli, J. (ed.) <u>Effective Schooling, Educational Change and the Role of the Principal</u>. Conference Proceedings. Northwest Regional Educational Laboratory, 1982.
- Persell, C.H. Education and Inequality: The Roots and Results of Stratification in America's Schools. New York: The Free Press, 1977.
- Pesseisen, B.Z. <u>Understanding Adolescence</u>: <u>Issues and Implications for Bffective Schools</u>. Philadelphia: Research for Better Schools, Inc., 1982.
- Peters, T.J. and R.H. Waterman Jr. In Search of Excellence: Lessons from American's Best-Run Companies. New York: Harper & Row Publishers, 1982.
- Purkey, S.C. and M.S. Smith. "Too Soon to Cheer? Synthesis of Research on Effective Schools." <u>Educational Leadership</u>. December, 1982, pp. 64-69.
- Research and Policy Committee of the Committee for Economic Development. (1985) Investing in Our Children: Business and The Public Schools.
- Richardson, E.L. "Employer Perceptions of the Preparation of Youth for Work." Paper presented at the Annual Meeting of the American Educational Research Association. Los Angeles, April 17, 1981.
- Rist, R.C. "Student Social Class and Teacher Expectations: The Self-Fulfilling Prohpecy in Ghetto Education." <u>Harvard Educational Review</u>. 1970, v40, pp. 411-451.
- Rivlin, A.M. Systematic Thinking for Social Action. Washington, D.C.: Brookings Institute, 1971.
- Rosenbaum, J.E. <u>Making Inequality: The Hidden Curriculum of High School</u> <u>Teaching</u>. New York: John Wiley & Sons, 1976.



Rosenshine, B. "Teaching Functions in Instructional Programs." Paper presented at the Airlee House National Invitational Conference on Research and Teaching: <u>Implications for Practice</u>. February 25, 1982.

The state of the s

Rutter, M. et al. <u>Fifteen Thousand Hours:</u> <u>Secondary Schools and Their Effects on Children</u>. Cambridge, Mass.: Harvard University Press, 1979.

Silberman, H. "Determining Goals for Vocational Education." Paper presented to the National Center for Research in Vocational Education. June, 1983.

Silberman, H. "High School Vocational Education: An Intrinsic Perspective.: Education. 1979, v21, pp. 46-51.

Sirotnik, K.A. "The Contextual Correlator of the Relative Expenditures of Classroom Time on Instruction and Behavior: An Exploratory Study of Secondary Schools and Classes." American Educational Research Journal. Summer, 1982, v19n2, pp. 275-292.

Sirotnik, K.A. "What You See is What You Get--Consistency, Persistency, and Mediocrity in Classroom." <u>Harvard Educational Review</u>. February, 1983, v53n2, pp. 16-31.

Slavin, R.E. "Cooperative Learning." Review of Educational Research. 1980a, v50n2, pp 315-342.

Slavin, R.E. <u>Using Student Team Learning: The Johns Hopkins Team Learning Project (Revised ed.)</u>. Baltimore: Center for Social Organization of Schools. The Johns Hopkins University, 1980b.

Slavin, R.E. <u>When Does Cooperative Learning Increase Student</u>

<u>Achievement?</u>

<u>Baltimore, Maryland: Center for Social Organization of Schools.</u>

The Johns Hopkins University, 1983.

Slavin, R.E. and Nancy L. Karweit. Mastery Learning and Student Teams.

A Component Building Experiment in Urban General Mathematics Classes.

Center for Social Organization of Schools. The Johns Hopkins University, 1983.

Stephens, J.M. The Influence of School Upon the Individual. Ann Arbor, Michigan: Edwards Brothers, 1933.

Stephens, J.M. The Process of Schooling. A psychological Examination. New York: Holt, Rinehait & Winston, 1967.

Swanson, G.I. "Policy Considerations in Achieving Excellence in Vocational Education." Presentation made at the American Vocational Association Annual Convention in Anaheim, CA, June, 1983.

Task Force on Education for Economic Growth. Action for Excellence. Education Commission of the States, June 1983, 50 pp.

Urban Superintendents Network, (1985, April). Towards Excellence: An Urban Response to the Recommendations for school Reform. New York: ERIC Clearinghouse on Urban Education.

Weinstein, R. "Reading Group Membership in First Grade: Teachers' Behaviors and Pupil Experience Over Time." Journal of Educational Psychology. 1976, v68, pp. 103-116.

1 7 4 Jun 1 38 4.

Weisberg, A. "What Research Has to Say About Vocational Education and the High Schools." Phi Delta Kappan, v14n5, January, 1983, pp. 355-359.

Wiley, D.E. "Another Hour, Another Day: Quantity of Schooling, A Potent Path for Policy,: In W.H. Sewell, R.M. Houser and D.L. Featherman (eds.), Schooling and Achievement in American Society. New York: Academic Press, 1976.

Wiley, D.E. and A. Harnischfeger. "Explosion of a Myth: Quantity of Schooling and Exposure to Instruction, Major Educational Variables." Educational Researcher. 1974, 3 (4), pp. 7-11.

Yin, R.K., Blank R.K. and White J.L. (1984) <u>Excellence in Urban High Schools: An Emerging District/School Perspective</u>. Washington D.C.: Cosmos Corporation.

Yin, R.K. (1985, June) <u>District/Secondary School Study: Interim</u>
Progress Report, Cosmos Update.